



# The New Zealand Gazette

WELLINGTON: THURSDAY, 13 SEPTEMBER 1990

### **Contents**

Vice Regal					 	 	3348
Parliamentary Summary	y				 	 	3348
Government Notices			٠.		 	 	3352
Authorities and Other A	Agenc	ies of S	State 1	Notices	 	 	3357
Land Notices					 	 	3358
Regulation Summary			٠.		 	 	3370

### Using the Gazette

The New Zealand Gazette, the official newspaper of the Government of New Zealand, is published weekly on Thursdays. Publishing time is 4 p.m.

Notices for publication and related correspondence should be addressed to:

Gazette Office, Department of Internal Affairs, P.O. Box 805, Wellington. Telephone (04) 738 699 Facsimile (04) 499 1865

or lodged at the Gazette Office, Seventh Floor, Dalmuir House, 114 The Terrace, Wellington.

Closing time for lodgment of notices at the Gazette Office: 12 noon on Tuesdays prior to publication (except for holiday periods when special advice of earlier closing times will be given).

Notices are accepted for publication in the next available issue, unless otherwise specified.

Notices being submitted for publication must be a reproduced copy of the original. Dates, proper names and signatures are to be shown clearly. A covering instruction setting out requirements must accompany all notices.

Copy will be returned unpublished if not submitted in accordance with these requirements.

#### Availability

The New Zealand Gazette is available on subscription from the Government Printing Office Publications Division or over the counter from Government Bookshops at:

Housing Corporation Building, 25 Rutland Street, Auckland.

33 Kings Street, Frankton, Hamilton.

25-27 Mercer Street, Wellington.

Mulgrave Street, Wellington.

E.S.T.V. House, 4185 Queens Drive, Lower Hutt.

159 Hereford Street, Christchurch.

Government Buildings, 1 George Street, Palmerston North.

Cargill House, 123 Princes Street, Dunedin.

#### Other issues of the Gazette:

Commercial Edition-Published weekly on Wednesdays.

Customs Edition-Published weekly on Tuesdays.

Special Editions and Supplements-Published as and when required.

### Vice Regal

### **Cabinet Office**

#### Resignation of the Prime Minister

His Excellency the Governor-General has been pleased to accept the resignation of

The Right Honourable Geoffrey Winston Russell Palmer of the Office of Prime Minister.

Dated at Wellington this 4th day of September 1990.

By Command:

MARIE SHROFF, Clerk of the Executive Council. vr9950

#### Appointment of Prime Minister

His Excellency the Governor-General has been pleased to appoint

The Right Honourable Michael Kenneth Moore to be Prime Minister.

Dated at Wellington this 4th day of September 1990.

By Command:

MARIE SHROFF, Clerk of the Executive Council.

### **Parliamentary Summary**

### **Bills Assented To**

#### Government Bills

Commodity Levies

7 September 1990 National Provident Fund Amendment

Contraception, Sterilisation and Abortion Amendment Postal Services Amendment

**Private Bill** 

7 September 1990

A. E. Thorpe Limited

Assent No.

126

127

128

129

3

### **Bills Introduced**

#### **Government Bills**

(Minister/Member in Charge Shown in Parenthesis)

4 September 1990

Orakei Bill. (Hon. K. T. Wetere.)

Health Commissioner Bill. (Rt. Hon. Helen Clark.)

Protected Areas (Prohibition on Mining) Bill. (Hon. Philip Woollaston.)

Trade in Endangered Species Amendment Bill. (Hon. Philip Woollaston.)

Social Welfare Bill (No. 2). (Hon. Dr. M. Cullen.)

Rehabilitation and Incapacity Bill. (Hon. Dr. M. Cullen.)

Driftnet Prohibition Bill. (Rt. Hon. Mike Moore.)

Harbour Boards Dry Land Endowment Revesting Bill. (Hon. Philip Woollaston.)

Companies Bill. (Hon. W. P. Jeffries.)

Courts Amendment Bill. (Hon. W. P. Jeffries.)

Proceeds of Crime Bill. (Hon. W. P. Jeffries.)

Dairy Board Amendment Bill. (Hon. Jim Sutton.)

Building Bill. (Hon. Margaret Austin.)

Museum of New Zealand Bill. (Hon. Margaret Austin.)

Human Rights Commission Amendment Bill. (Hon. W. P. Jeffries.)

#### Referred to Select Committee

Maori Affairs

Social Services

Planning and Development

Planning and Development

Social Services

Social Services

Foreign Affairs and Defence

Planning and Development

Justice and Law Reform

Justice and Law Reform

Justice and Law Reform

Primary Production

Internal Affairs and Local

Government

Internal Affairs and Local

Government

Justice and Law Reform

### **Summary of Bills Introduced**

Building Bill

The primary purpose of the Building Bill is to implement the major recommendations contained in the report of the Building Industry Commission.

The main aspects of the Bill are-

- (a) The introduction of a national focus for the control of building in New Zealand:
- (b) The establishment of a central agency (the Building Industry Authority) to act as one source of referral and review for buildingcontrol and occupancy-control systems within the framework of the National building code:
- (c) The detailing of distinct roles for the major policy and regulatory groups (the Minister of Internal Affairs, the Building Industry Authority, and territorial local authorities) in relation to building control and occupancy control:
- (d) With some minor exceptions, the binding of the Crown on building control and occupancy control matters:
- (e) The establishment of a legal basis for the making of a performance-based National building code to replace a variety of existing central government and local government building controls:
- (f) The introduction of provisions for the Building Industry Authority to undertake reviews of the operation of individual territorial authorities and building certifiers in relating to their control functions and providing for the Building Industry Authority to settle differences between property owners and territorial authorities relating to building control and occupancy control in terms of the National building code:
- (g) The introduction of a system of building and occupancy consents for new buildings and some existing buildings:
- (h) Making provision for the issue of certificates by the Building Industry Authority to building certifiers. A building certifier will be able to certify that building plans comply with the Building Act and code. A territorial authority will be required to accept such a certificate in its consideration of an application for building approval:
- (i) The introduction of an accreditation system to allow for innovation of new building products and processes:
- (j) The introduction of new procedures for the maintenance of buildings (other than dwelling units) and placing requirements on owners to ensure that standards are maintained in regard to those buildings.

#### Companies Bill

This Bill, intended when enacted to replace the Companies Act 1955, deals with the following matters:

- (a) The incorporation of companies.
- (b) Company names and name changes.
- (c) Company constitutions.
- (d) Reforms relating to shares.
- (e) Shareholders and their rights and obligations.
- (f) The meaning of "director" and "directors' powers and duties".
- (g) Measures relating to the enforcement of proceedings by or against companies.
- (h) The administration of companies.

- (i) Accounting records and financial statements.
- (j) Disclosure by companies by way of annual reports to shareholders and annual returns to the Registrar of Companies.
- (k) Amalgamation.
- (I) Compromises with creditors.
- (m) Liquidations.
- (n) Removal from the New Zealand Register of Companies.
- (o) Overseas companies and the registration of overseas companies as companies under this Act.
- (p) Registrar of Companies.
- (q) Offences and penalties.

#### Courts Amendment Rill

This Bill makes various changes to the jurisdiction of the Courts. In particular, the Bill does the following:

- (a) It establishes within the Court of Appeal a Criminal Appeal Division which will hear almost all criminal appeals to that Court:
- (b) It abolishes the Administrative Division of the High Court. In general, the jurisdiction presently exercised by that Division will be exercised by the High Court itself:
- (c) It increases the range of indictable offences that may be tried in a District Court that is presided over by a Judge appointed to exercise that jurisdiction under Part IIA of the District Courts Act 1947:
- (d) It removes the requirement for leave to appeal to the Court of Appeal in those criminal cases where leave is presently required:
- (e) It increases the general civil jurisdiction of District Courts:
- (f) It gives Family Courts jurisdiction concurrent with the High Court in proceedings under the Law Reform (Testamentary Promises) Act 1949 and the Family Protection Act 1955:
- (g) It gives Family Courts limited jurisdiction in the nature of wardship where it is necessary to protect the interests of the child pending the completion of complex or protracted proceedings relating to custody or access:
- (h) It gives Registrars of District Courts certain powers to transfer, or to adjourn, proceedings:
- (i) It requires at least 2 Judges of the High Court to hear and determine appeals against final decisions of a Family Court or of the Planning Tribunal, or questions stated by a Family Court or by that Tribunal.

#### Dairy Board Amendment Bill

The Bill alters the Board's constitution in relation to the appointment of directors and the exercise of its powers and privileges. The power of the Board to control the export of certain mixtures of dairy produce is removed, and in its place exporters can apply to the Board for permission to export independently. The power to make regulations regarding the marketing of bobby calfs and vells is also removed. Changes made by the Board in relation to the development of dairy calf and bobby calf industries and the development of the orderly marketing of bobby calfs and vells, will be subject to approval by the dairy industry. Miscellaneous provisions provide for the Board's financial statements to show additional information regarding its subsidiaries; statements to be audited by an Auditor appointed by the Board (not the Audit Office); efficiency reviews must be completed every 5 years; and removal of the Board from the control of the Higher Salaries Commission.

#### Driftnet Prohibition Bill

This Bill prohibits driftnet fishing and related activities in accordance with the requirements of the Convention for the Prohibition of Fishing with Long Driftnets in the South Pacific.

#### Harbour Boards Dry Land Endowment Revesting Bill

This Bill is a companion measure to the Foreshore and Seabed Endowment Revesting Bill and provides for certain former Harbour Board dry land endowments to be revested in the Crown for conservation or other purposes or held under the Reserves Act 1977.

#### Health Commissioner Bill

This Bill provides for the appointment of a Health Commissioner who will investigate complaints against persons or bodies that provide health care, and who will have general functions in relation to the protection of the rights of health consumers. The Bill also provides for the establishment of a Health Consumer Advocacy Service, and for the promulgation of a code of Health Consumers' Rights.

The Bill implements, in part, the recommendations contained in the "Report of the Committee of Inquiry into Allegations Concerning the Treatment of Cervical Cancer at National Women's Hospital and into Other Related Matters (1988)", otherwise known as the "Cartwright Report".

#### Human Rights Commission Amendment Bill

This Bill seeks to replace Part II of the Human Rights Commission Act 1977. Part II relates to unlawful discrimination. The number of grounds on which discrimination is prohibited is increased. New prohibited grounds of discrimination are: Health status; Age; Sexual orientation; Pregnancy; Political opinion; Trade Union involvement; Employment status; Family status; Identity of partner or relative

Each of the grounds specified in the new Part II of the principal Act is a prohibited ground of discrimination, for the purposes of that Act, if—

- (a) It pertains to a person or to a relative or associate of a person; and
- (b) It currently exists or has in the past existed; and

(c) It—

- (i) Exists in fact; or
- (ii) Is suspected or assumed or believed to exist by the person alleged to have discriminated.

The creation of new prohibited grounds of discrimination has required the creation of new exceptions. These new exceptions are set out in the new Part II.

#### Museum of New Zealand Bill

This Bill, which comes into force on 1 July 1991,—

- (a) Establishes the Museum of New Zealand; and
- (b) Constitutes the Museum of New Zealand Board, and defines the Board's functions and powers; and
- (c) Dissolves the Board of Trustees of the National Art Gallery, the National Museum, and the National War Memorial; and
- (d) Provides for the control and maintenance of the National War Memorial; and
- (e) Repeals the National Art Gallery, Museum, and War Memorial Act 1972.

The Museum of New Zealand is to be a repository for national treasures relating to our cultural heritage, history, and natural environment.

The new complex is to be located on the Wellington waterfront, and will incorporate the existing National Art Gallery and National Museum in Buckle Street, Wellington.

In place of the present Board of Trustees, the Bill constitutes the Museum of New Zealand Board, which will own the Museum site and administer the Museum. The functions and powers of the Board are defined in the body of the Bill, and the First Schedule details procedural and incidental matters relating to the Board.

The existing National War Memorial, and the land upon which it is situated, are to vest in the Crown. The Secretary for Internal Affairs will be responsible for the control and maintenance of the Memorial, and will administer the Memorial in conjunction with a council constituted by the Bill, to be known as the National War Memorial Advisory Council.

#### Orakei Bill

The Bill provides for the recognition of rights of the Ngati Whatua of Orakei in relation to the Orakei Block. It repeals the Orakei Block (Vesting and Use) Act 1978 and provides for the revesting of further portions of the Orakei Block in the Ngati Whatua o Orakei Maori Trust Board, and for the extension of the Trust Board's functions and powers.

The Bill sets aside other land as reserves, to be jointly administered by the Auckland City Council and the Trust Board, and protects land vested in the Trust Board by imposing certain restrictions on the disposal of land.

The Bill also grants exemptions to the Trust Board in relation to rates, reserves, contributions, land tax and other charges.

#### Proceeds of Crime Bill

This Bill makes provision for depriving persons convicted of serious offences of the proceeds of, and the benefits derived from, their offending.

The principal features of the Bill are as follows:

(a) The Bill provides for the commencement of confiscation proceedings in respect of a person who is convicted on indictment of an offence punishable by imprisonment for a term of 5 years or more. Special provision is made for persons who are charged with such an offence but who abscond before conviction.

Two kinds of order may be sought in confiscation proceedings. Forfeiture orders provide for the forfeiture to the Crown of identifiable property. Pecuniary penalty orders provide for the imposition of a monetary penalty as a measure of the benefits gained by a person from an offence.

In the context of drug-dealing offences, a special rule applies in respect of calculating the benefits derived from drug-dealing. All the property of the defendant at the time of the confiscation proceedings and any property that has passed through the defendant's hands in the previous 2 years is presumed to have been derived from the drug offences on which the proceedings are based. It is for the defendant to show that the property is not derived from that source:

- (b) The Bill provides for the issue of search warrants authorising the search for and seizure of property that may be liable to forfeiture under the Bill or may be required to satisfy a pecuniary penalty order. The Commissioner of Police is to hold seized property until it is dealt with under the Bill:
- (c) The Bill provides for the making of restraining orders in relation to property that may be subject to a forfeiture order or a pecuniary penalty order. The effect of a restraining order is that the property may not be disposed of or dealt with except as permitted by the Court.

The Official Assignee may be directed to administer the property while it remains subject to the restraining order, and where a forfeiture order or a pecuniary penalty order is made in respect of the property, to discharge that order out of the property:

(d) The Bill provides special powers to enable property gained from drug-dealing offences to be traced.

Production orders may be granted by a Court requiring any person to produce or make available to a police officer any document relevant to identifying, locating, or quantifying property that may be subject to confiscation proceedings under the Bill.

Monitoring orders may be granted by a Court requiring a financial institution to supply to the Police information about financial transactions conducted through that institution.

#### Protected Areas (Prohibition on Mining) Bill

This Bill prohibits mining in certain protected areas such as protected national parks and reserves, areas established under the Conservation Act 1987, the Wildlife Act 1953, the Reserves Act 1977 and in the Ross Dependency or Antarctica.

#### Rehabilitation and Incapacity Bill

This Bill relates to the provision of rehabilitation services in New Zealand and the system of benefits payable for incapacity.

Part I, Applications, deals with extent of cover, time limits, age limits, and general rights.

Part II contains provisions relating to rehabilitation programmes and conditions applied to rehabilitation, assessment, and reemployment of rehabilitated persons.

Part III, Weekly Benefits, provides for benefits in respect of earners and non-earners, provides for disability allowances, and sets out provisions relating to weekly earnings.

Part IV, Other Benefits, relates to the payment of funeral benefits, grants to surviving spouses, payment of benefits to surviving spouses and dependants; health care, attendant care, and household care costs.

Part V, Funding, relates to levies to be paid by employers, by the self-employed, and on motor vehicles, and provides for the funding of the Rehabilitation and Incapacity Corporation.

Parts VII and VIII set out provisions relating to applications, reviews and appeals.

Part IX contains miscellaneous provisions relating to benefits, such as overpayments and penalties, cancellation of benefits in certain circumstances and unclaimed benefits.

Part X concerns the relationship of benefits and other remedies.

Part XI constitutes the Accident Compensation Corporation as the Rehabilitation and Disability Corporation and defines its functions and composition.

Parts XII and XIV relate to transitional provisions and miscellaneous provisions.

#### Social Welfare Bill (No. 2)

This Bill is intended to consolidate and amend various Acts relating to social welfare and to provide for the introduction of a universal benefit.

Part I contains provisions relating to guaranteed retirement income and veterans' pensions, universal benefit, youth benefits, family assistance, accommodation benefit, special needs supplement, disability assistance, supplementary welfare programmes, lump sum payments on death, residential requirements, payment overseas of GRI and veterans' pensions, reciprocity agreements, fixing of rates of benefits, general provisions, provisions relating to overpayment of benefits, functions and powers of the Director-General, rights of review and appeal, and regulations.

Part II relates to liable parent contributions.

Part III contains provisions relating to community welfare and disability assistance services.

Part IV re-enacts, with minor amendments, those provisions of the Disabled Persons Community Welfare Act 1975 relating to the registration of homes for people with disabilities.

Part V re-enacts, with minor amendments, those provisions of the Disabled Persons Community Welfare Act 1975 relating to access to premises for people with disabilities.

Part VI re-enacts the current provisions relating to the Social Welfare Commission, District Executive Committees and Area Welfare Executive Committees, advisory committees, district review committees, the Social Security Appeal Authority, the New Zealand Artificial Limb Board, and the delegation of functions and powers by the Director-General.

Part VII contains miscellaneous and transitional provisions.

Part VIII provides for consequential amendments to the Income Tax Act 1976.

#### Trade in Endangered Species Amendment Bill

This Bill increases significantly the penalties for offences against the Trade in Endangered Species Act 1989 and insets into that Act new enforcement provisions, including the power to arrest without a warrant.

ps10084

### **Government Notices**

### **Agriculture and Fisheries**

#### **Noxious Plants Act 1978**

Notice Declaring Sagittaria (Sagittaria graminea ssp. platyphylla) a Class B Noxious Plant (No. 5032; Ag. 1570-03-3)

1. Pursuant to section 19 of the Noxious Plants Act 1978, the Noxious Plants Council hereby declares Sagittaria (Sagittaria

graminea ssp. platyphylla) to be a Class B noxious plant throughout New Zealand.

2. This notice shall come into effect on the day after its date of notification in the *Gazette*.

Dated at Wellington this 6th day of September 1990.

J. L. RANDALL, Secretary, Noxious Plants Council. go10091

### **Education**

#### **Education Amendment Act 1990**

#### University of Canterbury Notice 1990

Pursuant to section 168 of the Education Amendment Act 1990, the Minister of Education gives the following notice.

#### Notice

- 1. (a) This notice may be cited as the University of Canterbury notice 1990.
- (b) This notice shall come into force on the date of its publication in the Gazette.
- 2. There shall be a Council to be known as the University of Canterbury Council which shall control the University of Canterbury.
- 3. The University of Canterbury Council shall be constituted as follows:
  - (a) Four members appointed by the Minister of Education.
  - (b) The Vice-Chancellor of the University of Canterbury.
- (c) One permanent member of the academic staff of the University of Canterbury elected by the permanent members of that staff.
- (d) Two academic staff members of the Academic Board of the University of Canterbury elected by the academic staff members of that Board.
- (e) One permanent member of the general staff of the University of Canterbury elected by the permanent members of that staff.
- (f) Two members who are or have been students of the University of Canterbury appointed by the Executive of the University of Canterbury Students' Association Incorporated.
- (g) One member appointed by the Council after consultation with the central organisation of employers within the meaning of the Labour Relations Act 1987.
- (h) One member appointed by the Council after consultation with the central organisation of workers within the meaning of the Labour Relations Act 1987.
- (i) Three members co-opted by the University of Canterbury Council.
- (j) Four graduates elected by the Court of Convocation of the University of Canterbury.
- 4. The term of office of members of the Council, vacation of office, disclosure of members' interest and casual vacancies are covered in section 173–176 of the Education Amendment Act 1990

Subject to section 173, the maximum term for any member, other than the Vice-Chancellor, shall be three consecutive 4 year terms.

5. The powers of the University of Canterbury Council shall not be affected by vacancy in the membership thereof.

Dated at Wellington this 12th day of September 1990.

PHIL GOFF, Minister of Education. go10104

#### Victoria University of Wellington Notice 1990

Pursuant to section 168 of the Education Amendment Act 1990, the Minister of Education gives the following notice.

#### Notice

- 1. (a) This notice may be cited as the University of Victoria notice 1990.
- (b) This notice shall come into force on the date of its publication in the Gazette.
- 2. There shall be a Council to be known as the Victoria

- University of Wellington Council which shall control the Victoria University of Wellington.
- 3. The Victoria University of Wellington Council shall be constituted as follows:
  - (a) Four persons appointed by the Minister of Education.
  - (b) The Vice-Chancellor of Victoria of Wellington.
- (c) Three permanent members of the academic staff of the Victoria University of Wellington, one at least of whom shall be a Professor of the University and one at least of whom shall not be a Professor, elected by the permanent members of that staff in accordance with the VUW Council Elections Statute 1990.
- (d) One permanent member of the general staff of Victoria University of Wellington elected by the permanent members of that staff in accordance with the VUW Council Elections Statute 1990.
- (e) Two persons who are or have been students of Victoria University of Wellington appointed by the Executive for the time being of the Victoria University of Wellington Students' Association
- (f) One person appointed by the Council of Victoria University of Wellington after consultation with the New Zealand Employers' Federation.
- (g) One person appointed by the Council of Victoria University of Wellington after consultation with the New Zealand Council of Trade Unions.
- (h) Five persons being graduates of the University of New Zealand or of any University in New Zealand, to be elected by the Court of Convocation for the University in accordance with the VUW Council Elections Statute 1990.
- (i) Two persons co-opted by the Council of Victoria University of Wellington.
- 4. The term of office of members of the Council, vacation of office, disclosure of interest by members and casual vacancies are covered in section 173–176 of the Education Amendment Act 1990.

Subject to section 173, the maximum term for any member, other than the Vice-Chancellor, shall be five consecutive 4 year terms.

5. The powers of the Victoria University of Wellington Council shall not be affected by vacancy in the membership thereof.

Dated at Wellington this 12th day of September 1990.

PHIL GOFF, Minister of Education. go10105

#### Otago Polytechnic Notice 1990

Pursuant to section 168 of the Education Amendment Act 1990, the Minister of Education gives the following notice.

#### Notice

5

- 1. (a) This notice may be cited as the Otago Polytechnic notice 1990.
- (b) This notice shall come into force on the date of its publication in the *Gazette*.
- 2. There shall be a Council to be known as the Otago Polytechnic Council which shall control the Otago Polytechnic.
- 3. The Otago Polytechnic Council shall be constituted as follows:
  - (a) Four members appointed by the Minister of Education.
  - (b) The Chief Executive Officer of Otago Polytechnic.
- (c) One permanent member of the academic staff of Otago Polytechnic elected by the permanent members of that staff.
- (d) One permanent member of the general staff of Otago Polytechnic elected by the permanent members of that staff.
  - (e) One member who is or has been a student of Otago

Polytechnic who shall represent the students of Otago Polytechnic and who shall be elected by the students of Otago Polytechnic.

- (f) One member appointed by the Otago Polytechnic Council after consultation with the New Zealand Employers' Federation.
- (g) One member appointed by the Otago Polytechnic Council after consultation with the New Zealand Council of Trade Unions.
- (h) One member, being a member of the Dunedin City Council, to be appointed by that Council.
- (i) One member being a member of the Council of the University of Otago, or, of the Senate of that University, to be appointed by the Council of the University of Otago.
- (j) One member being a member of the Dunedin College of Education Council, to be appointed by that Council.
- (k) One member to be appointed by the Otago District Maori Executive.
- (I) One member to be appointed by an electoral college comprising the local branches of Federated Farmers of New Zealand, Country Women's Institute, Women's Division of Federated Farmers of New Zealand, Young Farmers Club, and the Fruitgrowers' Federation of New Zealand.
- (m) Up to 5 members to be co-opted by the Otago Polytechnic Council if and when it thinks fit taking into account the specific needs of the Otago Polytechnic, its unique character, and the gender, ethnic and socio-economic composition of the Otago community.
- 4. The term of office of members of the Council, vacation of office, disclosure of members' interest and casual vacancies are covered in section 173–176 of the Education Amendment Act 1990.
- 5. Subject to section 173, the maximum term for any member, other than the Chief Executive Officer, shall be three consecutive 4 year terms.

The powers of the Otago Polytechnic Council shall not be affected by vacancy in the membership thereof.

Dated at Wellington this 12th day of September 1990.

PHIL GOFF, Minister of Education. go10106

# Private Schools Conditional Integration Act 1975

# Attendance Dues for Michael Park School, Auckland

Pursuant to section 36 of the Private Schools Conditional Integration Act 1975, notice is given that attendance dues of \$240 per term have been approved by the Minister of Education.

Dated at Wellington this 20th day of August 1990.

C. GIBSON, Group Manager, National Operations. go10077

#### Private Schools Conditional Integration Act 1975

Pursuant to section 10 of the Private Schools Conditional Integration Act 1975, notice is given that a supplementary integration agreement has been signed between the Minister of Education on behalf of Her Majesty the Queen acting through the Manager, District Operations, Central West Wanganui, Ministry of Education pursuant to delegated authority, and St Mary's Diocesan School (Incorporated) as proprietor of the following school:

#### St Mary's Diocesan School for Girls, Stratford

The said supplementary integration agreement was executed on the 24th day of August 1990. Copies of the supplementary integration agreement are available for inspection without charge by any member of the public at the District Office of the Ministry of Education.

D. BURNEY, for Secretary of Education. go10082

### Energy

#### **Coal Mines Act 1979**

#### Member of the Board of Examiners Appointed

Pursuant to section 210 of the Coal Mines Act 1979, the Minister of Energy hereby appoints

John Martin Stark

to be a member of the Board of Examiners from 1 May 1990, for a period of 3 years.

Dated at Wellington this 20th day of August 1990.

D. BUTCHER, Minister of Energy. go10080

#### Member of the Board of Examiners Appointed

Pursuant to section 210 of the Coal Mines Act 1979, the Minister of Energy hereby appoints  ${\sf Min}$ 

Dennis John Wood

to be a member of the Board of Examiners from 1 May 1990, for a period of 3 years.

Dated at Wellington this 20th day of August 1990.

D. BUTCHER, Minister of Energy. go10081

Proprietary Name

(if any)

### Health

#### **Medicines Act 1981**

#### Consent to the Distribution of a New Medicine

Pursuant to section 20 of the Medicines Act 1981, the Minister of Health hereby consents to the distribution in New Zealand of the new medicine set out in the Schedule hereto:

#### Schedule

Name and Strength

Piroxicam 10 mg, 20 mg

Form Capsule

Name and Address of Manufacturer

Glaxo New Zealand Ltd., Palmerston

North, New Zealand

This replaces the consent gazetted 30 August 1990.

Dated this 8th day of September 1990.

HELEN CLARK, Minister of Health. qo10085

### **Justice**

#### **Disputes Tribunal Act 1988**

#### Disputes Tribunals Referee Appointed

Pursuant to section 7 of the Disputes Tribunals Act 1988, His Excellency the Governor-General has been pleased to appoint

Josephine Mary O'Donnell

of Auckland, to be a Referee to exercise the jurisdiction of the Disputes Tribunals for a term of 3 years on and from the date hereof.

Dated at Wellington this 4th day of September 1990.

P. F. DUNNE, for Minister of Justice.

#### Disputes Tribunals Referee Appointed

Pursuant to section 7 of the Disputes Tribunals Act 1988, His Excellency the Governor-General has been pleased to appoint

Rosemary Petherick

Justice of the Peace of Wanganui, to be a Referee to exercise the jurisdiction of the Disputes Tribunals for a term of 3 years on and from the date hereof.

Dated at Wellington this 6th day of September 1990.

P. F. DUNNE, for Minister of Justice. go10115

#### Marriage Act 1955

#### Marriage Approval of Organisations, Notice No. 10

Pursuant to the Marriage Act 1955, the Registrar-General of Marriages, hereby gives notice as follows:

#### Notice

- 1. This notice may be cited as the Marriage (Approval of Organisations) Notice No. 10.
- 2. The organisations specified in the Schedule hereto are hereby declared to be approved organisations for the purpose of the Marriage Act 1955.

#### Schedule

Piako Christian Fellowship.

Pinehaven Fellowship.

The Victory Christian Ministries.

Witness in the Word of Jesus Christ.

Dated at Lower Hutt this 11th day of September 1990.

B. E. CLARKE, Registrar-General go10107

#### Marriage Celebrants for 1990, Notice No. 16

Pursuant to the provisions of section 13 of the Marriage Act 1955, it is hereby notified that the following names have been removed from the list of Marriage Celebrants under section 10 of the Act:

Burson, Gregory Laine, Baptist.

Lawerence, Frank John, Anglican.

Long, William John, Jehovah's Witnesses.

Raynor, Colin Robert, Elim Church of New Zealand.

Rikihana, Hamahona, Ringatu.

Stratford, Melvin Douglas, Elim Church of New Zealand.

Tasker, William George, Church of Jesus Christ of Latter-Day Saints.

Wilson, Jack, Presbyterian.

Dated at Lower Hutt this 11th day of September 1990.

B. E. CLARKE, Registrar-General. go10108

#### Marriage Celebrants for 1990, Notice No. 17

Pursuant to the provisions of section 10 of the Marriage Act 1955, the following names of Marriage Celebrants within the meaning of the said Act are published for general information.

Adolph, Arthur William Henry, Apostolic Church.

Bowker, Noel William, Apostolic Church.

Breukelaar, Gerrit Willem, Jehovah's Witnesses.

Brown, Russell Gurney, Apostolic Church.

Burson, Gregory Laine, Harvest Christian Centre.

Cochran, Brian Rober, Baptist.

Conrad, Jay Dee, Presbyterian.

Crozier, John Peart, Presbyterian.

Dodd, Alastair Jenkins, Brethren.

Edgar, David, Anglican.

Edwardson, Mita Thomas, Gisborne Christian Fellowship.

Faletau, Siosateki Tona, Assemblies of God in NZ.

Graham, Stanley Hugh, Christadelphians.

Hawkins, Philip James, Jehovah's Witnesses.

Hilton, Russell James, Brethren.

Universal Fellowship of Michael Leonard, Metropolitan Community Churches in NZ Trust Board.

Hukui, Don Maka, Jehovah's Witnesses.

Hunt, Gordon Patrick, Church of Jesus Christ of Latter-Day Saints.

Hussain, Manzur, New Zealand Muslim Association.

Johnston, Ian Mark, Presbyterian.

Lawrence, Frank John, Anglican.

Livingstone, William John Allen, Garden City Church.

Love, Rawiri Powhiri, New Wine Fellowship.

MacAnn, Rodney George, Baptist.

McFarlane, Noel Hilton, Salvation Army.

McKinlay, Judith Elizabeth, Presbyterian.

Moyle, John Jeffrey, Baptist.

Pohahau, Senituli, Vision Ministries.

Pratt, Peter, Jehovah's Witnesses.

Rayner, Colin Robert, Elim Church of New Zealand.

Renner, Peter John, Baptist.

Roberts, Colin Lloyd, Christadelphians.

Robinson, John, Apostolic Church.

Scheirer, Francis, Presbyterian.

Shepherd, Owen Collis, Apostolic Church.

Short, Harold Frederick, Garden City Church.

Sipaia, Mulinuu Lesa, First Samoan Full Gospel Pentecostal Church Incoporated NZ

Struthers, Neil Campbell, Garden City Church.

Tuck, Maxwell Henry, Hastings Christian Community Church.

Tyler-Whiteman, Ross Norman, Vision Ministries.

Walker, Dennis John, Baptist.

Winther, John Herbert, Vision Ministries.

Wolfe, Tony Kester, Methodist.

Dated at Lower Hutt this 11th day of September 1990.

B. E. CLARKE, Registrar-General. go10109

### Labour

#### **Labour Relations Act 1987**

#### Cancellation of Registration of Employers Organisation for Failure to Make Annual Return

Pursuant to section 32 (3) of the Labour Relations Act 1987, it is hereby notified that the registration of the Wellington Furniture and Furnishing Trade Organisation of Employers, registered No. 2/14, situated at Wellington has been cancelled as from the following date for failure to deliver to the Registrar a full and correct return in accordance with section 52 of this Act

Dated at Wellington, this 10th day of September 1990.

M. E. FEELY, Registrar of Unions, Department of Labour. 6 go10110

### **Survey and Land Information**

#### **Surveyors Professional Regulations** 1977

#### Survey Board of New Zealand

Pursuant to regulation 9 (2) of the Surveyors Professional Regulations 1977, the following notice is hereby published:

#### **Surveyors Examination**

The New Zealand Acts and Regulations written examination will be held on Friday, the 16th day of November 1990 at  $9\,\mathrm{a.m.}$ 

Applications, on the proper form together with the prescribed fee, must reach the secretary of the board before 30 October 1990.

Application forms and information about the board's requirements should be obtained from the undersigned.

The prescribed fee for the examination is \$33.75 incl. GST.

D. R. RINCKES, Secretary.

Survey Board of New Zealand, care of Department of Survey and Land Information, C.P.O. Box 170, Wellington. 909944

### **Transit New Zealand**

#### **Transit New Zealand Act 1989**

#### Corrigendum

# Declaring State Highway to be a Limited Access Road State Highway No. 2

In the notice with the above heading published in the *New Zealand Gazette*, 19 April 1990, No. 60, page 1337, lines 5, 6 and 7 delete "Lots 2, 3, 6 and 4 of D.P. S. 3875, being in certificates of title 19A/254, 19A/255, 19A/175 and 19A/174" and replace with "Lots 1, 2, 3 and 4, D.P. S. 3875 and D.P. S. 55509, being in certificates of title 46A/95, 46A/96, 46A/97 and 19A/174."

Dated at Wellington this 4th day of September 1990.

M. K. LAUDER, State Highways Manager. go10078

### **Transport**

#### **Traffic Regulations 1976**

# Approval of Motorcycle Safety Helmets in Terms of the Traffic Regulations 1976

Pursuant to subclause (1) of regulation 88 of the Traffic Regulations 1976\* and pursuant to powers delegated to me by the Secretary for Transport by an instrument of delegation dated the 14th day of November 1989, motorcycle safety helmets of the make and type described in the Schedule hereto are hereby approved for the purposes of regulation 31 of the said regulations.

#### Schedule

(1) Safety helmets manufactured by Arai Helmet Ltd., Japan, complying with E.C.E. Regulation 22, with the following type designations and bearing the marks of homologation:

Arai MX80, MX-V, MX-PRO 1 and 2, Urban, Sinisalo E4 031003 —

Arai MX80, MX-V, MX-Pro 1 and 2, Urban, Sinisalo E4 031004 —

Arai MX80, MX-V, MX-Pro 1 and 2, Urban, Sinisalo E4 031005 —

all with required serial numbers.

Dated at Wellington this 5ty day of September 1990.

H. C. MATHESON, Senior Automotive Engineer.

\*S.R. 1976/227 Traffic Regulations 1976.

(14/42/22) go10116

# Transport (Vehicle and Driver Registration and Licensing) Act 1986

#### Approval of Hazardous Substances Courses

Pursuant to paragraph 48 (2) (e) of the Transport (Vehicle and Driver Registration and Licensing) Act 1986; and

Under powers delegated to me by the Secretary for Transport in an instrument of delegation dated the 14th day of November 1989:

I, John Battersby Toomath, Manager Safety Standards, hereby approve the following courses and organisations for the purposes of paragraph 70 H (1) (a) of the Transport Act 1962:

TNL Group Ltd. Hazardous Substances Course conducted by TNL Group Ltd. as the organisation.

Signed at Wellington this 11th day of September 1990.

J. B. TOOMATH, Manager Safety Standards.

(File: 16/6/1/5) go10083

#### Approval of Defensive Driving Courses

Pursuant to section 48 (2) (a) of the Transport (Vehicle and Driver Registration and Licensing) Act 1986, and pursuant to a delegation given by the Secretary for Transport in an instrument dated the 14th day of November 1989; I, Warren Richard Gillespie Simeon, Controller, Traffic Education Standards, hereby approve the following organisations for the purposes of section 68 of the Transport Act 1962 and Regulations 11 (1) (b) (i) and 32 (2) of the Transport (Drivers Licensing) Regulations 1987;

Able Tasman Defensive Driving Course, Motueka.

J. A. Simpson Defensive Driving Course, Hamilton.

Signed at Wellington this 10th day of September 1990.

W. R. G. SIMEON, Controller, Traffic Education Standards.

(File: 16/6/1/2) go10111

#### **Approval of Traffic Improvement Schools**

Pursuant to section 48 (2) (a) of the Transport (Vehicle and Driver Registration and Licensing) Act 1986, and pursuant to a delegation given by the Secretary for Transport in an instrument dated the 14th day of November 1989; I, Warren Richard Gillespie Simeon, Controller, Traffic Education Standards, hereby approve;

As organisations or instructors for the purpose of conducting courses pursuant to regulations 11 (1) (b) (i) and 32 (2) of the Transport (Drivers Licensing) Regulations 1987.

Carolyn B. Evans Traffic Improvement School, Christchurch; under the direction of Carolyn B. Evans.

Les Stone Traffic Improvement School, Whangarei; under the direction of Les Stone.

Jason Goile Traffic Improvement School, Whangarei; under the direction of Jason Goile.

Richard Evans Traffic Improvement School, Whangarei; under the direction of Richard Evans.

Jeanette Evans Traffic Improvements School, Whangarei; under the direction of Jeanette Evans.

Signed at Wellington this 10th day of September 1990.

W. R. G. SIMEON, Controller, Traffic Education Standards.

(File: 16/6/5/1) go10112

#### Approval of Course in Map Reading

Pursuant to section 48 (2) (a) of the Transport (Vehicle and Driver Registration and Licensing) Act 1986; and pursuant to a delegation given by the Secretary for Transport in an instrument dated the 14th day of November 1989, I, Robert Albert Gibson, National Co-ordinator Driver Licensing, hereby approve the following organisation to conduct courses in map reading for the purposes of regulation 11 (1) (c) of the Transport (Driver Licensing) Regulations 1987:

Willis Street Driving School, Papakowhai.

Signed at Wellington this 7th day of September 1990.

R. A. GIBSON, National Co-ordinator Driver Licensing.

(File: 16/6/1/4) go10113

# **Authorities and Other Agencies of State**

### **PostBank**

### Post Office Savings Bank Regulations 1985

Bonus Bonds Weekly Prize Draw No. 2, 8 September 1990

Pursuant to the Post Office Savings Bank Regulations 1985, notice is hereby given that the result of the weekly Prize Draw No. 2 for 8 September is as follows:

One prize of \$50,000: 883 764808.

Twenty-five prizes of \$5,000: 214 955892, 821 824635, 992 657981, 1072 982922, 1073 265100, 1223 751590, 1422 302459, 2110 497775, 2198 856510, 2511 829245, 2617 275308, 2827 203850, 2914 675905, 3527 773061, 4095 247307, 4628 299586, 4720 388789, 5027 835948, 5894 436002, 6397 786946, 8494 193231, 9393 217228, 9683 405374, 9990 919545 and 9996 048295.

DAVID CAYGILL, Minister of Finance. au9952

### **Land Notices**

### Conservation

#### Land Act 1948

#### Reservation of Land

Pursuant to the Land Act 1948, the Minister of Conservation with the consent of the Minister of Lands, hereby sets apart the land described in the Schedule hereto, as a local purpose (dog pound site) reserve, subject to the provisions of the Reserves Act 1977.

#### Schedule

#### South Auckland Land District—Kawerau District

4118 square metres, more or less, being Allotment 1364, Matata Parish, situated in N.Z.M.S. 261 V15 Sheet 6.4. Part New Zealand Gazette, 1884, page 237. S.O. Plan 56881.

Dated at Wellington this 3rd day of September 1990.

PHILIP WOOLLASTON, Minister of Conservation.

(DOC H.O. ORE0039 : C.O. 7/78/2) ln9949

9949

#### Reserves Act 1977

#### Classification of Reserve

Pursuant to the Reserves Act 1977, and to a delegation from the Minister of Conservation, the Manager, Protection Division, Canterbury, hereby classifies the reserve, described in the Schedule hereto, as an historic reserve, subject to the provisions of the said Act.

#### **Schedule**

#### Canterbury Land District—Christchurch City

#### Coronation Hill Reserve

3.4170 hectares, more or less, being Reserve 3900, situated in Block III, Halswell Survey District. Part certificate of title 275/9. New Zealand Gazette, 1912, page 3352. Subject to section 6, Summit Road (Canterbury) Protection Act. Document 22254/1. S.O. Plan 4861.

Dated at Christchurch this 31st day of August 1990.

K. F. D. HUGHEY, Manager, Protection Division.

(DOC H.O.; C.O. 1/20/3/1)

#### Revocation of the Reservation Over a Reserve Specifying the Manner of Disposal and How Proceeds of Sale Shall be Utilised

Pursuant to the Reserves Act 1977, and to a delegation from the Minister of Conservation, the Regional Conservator, Canterbury, hereby revokes the reservation as a local purpose (road) reserve over the land, described in the Schedule hereto, and further, declares that the said land may be disposed of by the Christchurch City Council in such manner, at such price and on such terms and conditions as the council shall determine, the proceeds from any such sale to be paid into the council's reserves account.

#### Schedule

#### Canterbury Land District—Christchurch City

536 square metres, more or less, being Lot 4, D.P. 13196, situated in Block VII, Christchurch Survey District. Local Purpose (Road) Reserve by section 6, Counties Amendment Act 1972.

Dated at Christchurch this 31st day of August 1990.

M. J. CUDDIHY, Regional Conservator.

(DOC H.O.; C.O. 1/20/4/1)

ln9946

### Revocation of Appointment to Control and Manage a Reserve

Pursuant to the Reserves Act 1977 and to a delegation from the Minister of Conservation, the Regional Conservator, Waikato Conservancy, Department of Conservation, hereby revokes the appointment of the Waitomo District Council to control and manage the recreation reserve, described in the Schedule hereto.

#### Schedule

#### South Auckland Land District—Waitomo District

3.6270 hectares, more or less, being Section 28, Block X, Orahiri Survey District. All *New Zealand Gazette*, 1983, page 1385, document H465147. S.O. Plan 41658.

Dated at Hamilton this 5th day of September 1990.

G. E. ROWAN, Regional Conservator.

(DOC Ref. R.S.C. 178, R.R.L. 032)

in9947

Revocation of Appointment of the Minister of Internal Affairs to Control and Manage a Reserve

Pursuant to the Reserves Act 1977, the Minister of Conservation hereby revokes the appointment of the Minister of Internal Affairs to control and management of Tuckers Beach Wildlife Management Reserve, described in the Schedule hereto.

#### Schedule

#### Otago Land District

150.835 hectares, more or less, being Section 92, Block II, and Section 85, Block IV, Shotover Survey District. *New Zealand Gazette*, 1975, page 1863. S.O. Plan 18180.

Dated at Wellington this 3rd day of September 1990.

PHILIP WOOLLASTON, Minister of Conservation.

(DOC H.O. Res. 12/6/11: C.O. CML. 16/W/9)

# Revocation of the Reservation Over Part of Reserves

Pursuant to the Reserves Act 1977, and to a delegation from the Minister of Conservation, the Regional Conservator, Waikato Conservancy, Department of Conservation, hereby revokes the reservation over the land, described in the Schedule hereto, as reserves as specified at the end of the respective descriptions of the said lands.

#### Schedule

#### South Auckland Land District—Waitomo District

1.2141 hectares, more or less, being Hauturu East No. 1A6 Block, situated in Block X, Orahiri Survey District. All *New Zealand Gazette*, 1906, page 11. S.O. Plan 38686. Scenic Reserve.

3.6270 hectares, more or less, being Section 28, Block X, Orahiri Survey District. All *New Zealand Gazette*, 1983, page 1385, document H.465147. S.O. Plan 41658. Recreation Reserve.

1438 square metres, more or less, being part Lot 1, D.P. 30223 and part Section 12, Block X, Orahiri Survey District.

2

1CL

All New Zealand Gazette, 1983, page 495, document H.455743. S.O. Plan 51020. Local Purpose Reserve (site for a museum).

Dated at Hamilton this 5th day of September 1990.

G. E. ROWAN, Regional Conservator.

(DOC Ref. R.S.C. 178, R.R.L. 032, LES 202)

1CL

# Appointment of Banks Peninsula District Council to Control and Manage and Naming of a Reserve

Pursuant to the Reserves Act 1977, and to a delegation from the Minister of Conservation, the Regional Manager hereby appoints the Banks Peninsula District to control and manage the reserve described in the Schedule hereto, subject to the provisions of the said Act and an agreement dated the 28th day of August 1990, as a reserve for historic purposes, and further declares that the said reserve shall hereafter be known as Akaroa Court House Historic Reserve.

#### Schedule

#### Canterbury Land District—Akaroa County

1051 square metres, more or less, being Rural Section 41440, situated in Block IV, Akaroa Survey District. All *New Zealand Gazette*, 1982, page 188. S.O. Plan 13420.

Dated at Christchurch this 3rd day of September 1990.

M. CUDDIHY, Regional Manager.

(DOC R.O. 3/6/1)

2

### **Justice**

#### **Maori Affairs Act 1953**

# Setting Apart Maori Freehold Land as a Maori Reservation

Pursuant to section 439 of the Maori Affairs Act 1953, the Maori freehold land described in the Schedule hereto is hereby set apart as a Maori reservation for the purpose of a historical reserve and wahi tapu for the common use and benefit of the Whanau a Kahu, Whanau a Tuwhakiriora and Ngati-Rangi.

#### Schedule

#### Gisborne Land District

All that piece of land situated in Block XIII, East Cape Survey District and described as follows:

Area

ha Being

22.8268 Marangairoa A27, and being all the land contained in the consolidation order, dated 22 March 1957.

Dated at Wellington this 21st day of August 1990.

W. GARDINER, General Manager, Iwi Transition Agency.

(D.O. Appln. No. 90679; H.O. MLC 2/3/1/5) 2CL ln10093

#### Cancelling the Reservation of Land and Setting Apart Maori Freehold Land as a Maori Reservation

Pursuant to section 439 (5) of the Maori Affairs Act 1953, the reservation known as part Whangawehi IB3D2B3 constituted by notice, dated the 29th day of May 1978 and published in the *New Zealand Gazette*, No. 50, page 1619 of 8 June 1978, is hereby cancelled; and notice is hereby given that pursuant to section 439 of the Maori Affairs Act 1953, the Maori freehold

land described in the Schedule hereto, is hereby set apart as a Maori reservation for the purpose of a graveyard for the common use and benefit of the Greening family.

#### Schedule

#### Hawke's Bay Land District

All that piece of land situated in Block II, Mahia Survey District and described as follows:

Area m²

Being

632 Part Whangawehi 1B3H and being part of the land contained in the partition order, dated 27 November 1984.

Dated at Wellington this 21st day of August 1990.

W. GARDINER, General Manager, Iwi Transition Agency.

(D.O. 8/3/528; H.O. MLC 2/3/1/5) ln10095

2CL

#### Setting Apart Maori Freehold Land as a Maori Reservation

Pursuant to section 439 of the Maori Affairs Act 1953, the Maori freehold land described in the Schedule hereto is hereby set apart as a Maori reservation for the purpose of a marae for the common use and benefit of the Ngati Hinehika hapu of Te Reinga.

#### Schedule

#### Gisborne Land District

All that piece of land situated in Block II, Opoiti Survey District and described as follows:

Area m²

.5960 Te Reinga 2D5B1 and being all the land contained in the vesting order, dated 3 July 1988.

Being

Dated at Wellington this 21st day of August 1990.

W. GARDINER, General Manager, Iwi Transition Agency.

(D.O. Appln. No. 89990; H.O. MLC 2/3/1/5) h10098

2CL

### Setting Apart Maori Freehold Land as a Maori Reservation

Pursuant to section 439 of the Maori Affairs Act 1953, the Maori freehold land described in the Schedule hereto is hereby set apart as a Maori reservation for the purpose of a marae for the common use and benefit of the Whanau a Kahu, Whanau a Ruataupare and Ngati-Rangi.

#### Schedule

#### Gisborne Land District

All that piece of land situated in Block VIII, Tokomaru Survey District and described as follows:

Area

m<sup>2</sup> Being

167 Part Section 4B, Block II, Tuatini Maori Township and being all the land comprised and described in certificate of title 2C/1218.

Dated at Wellington this 21st day of August 1990.

W. GARDINER, General Manager, Iwi Transition Agency.

(D.O. Appln. No. 90680; H.O. MLC 2/3/1/5) ln10099

2CL

# New Zealand Railways Corporation

# New Zealand Railways Corporation Act 1981

# Declaring Land at Pukekohe to be Set Apart for Railway Purposes

Pursuant to sections 24 and 30 of the New Zealand Railways Corporation Act 1981 and section 52 of the Public Works Act 1981, the Chief Executive of the New Zealand Railways Corporation hereby declares the land described in the Schedule hereto to be set apart for and on behalf of Her Majesty the Queen for railway purposes.

#### Schedule

#### North Auckland Land District—Franklin District

All that piece of land containing 111 square metres (4.4p), situated in Block XV, Drury Survey District, being part Allotments 12 and 16, Pukekohe Parish. All conveyance 244329 (D.I. 18A/44).

Dated at Wellington this 6th day of September 1990.

P. K. TROTMAN, for Chief Executive, New Zealand Railways Corporation.

(NZR L.O. 6961/320) ln10101

### **Survey and Land Information**

#### **Public Works Act 1981**

#### Land Acquired for Road, Land Declared to be Road, Road Stopped and Land Taken in the District of Rangitikei

Pursuant to the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Manager, Department of Survey and Land Information, Wellington:

- (a) Pursuant to section 20 (1), declares that agreements to that effect having been entered into, the land described in the First Schedule hereto is hereby acquired for road and vested in The Rangitikei District Council on the date of publication hereof in the *Gazette*.
- (b) Pursuant to section 114 (1), declares the land described in the Second Schedule hereto to be road and vested in The Rangitikei District Council on the date of publication in the Gazette.
- (c) Pursuant to sections 116 (1), 117 and 120 (3), declares the portions of road described in the Third, Fourth, Fifth, Sixth, Seventh, Eighth and Ninth Schedules hereto to be stopped and declares that:
  - (i) The areas described in the Third Schedule shall be amalgamated with the land in certificate of title, No. 29D/892, subject to memorandum of mortgage 916768.13.
  - (ii) The areas described in the Fourth Schedule shall be amalgamated with the land in certificate of title, Volume 170, folio 243, subject to memoranda of mortgage 906006.7, 906006.8 and 186733.3.
  - (iii) The area described in the Fifth Schedule shall be amalgamated with the land in certificate of title, Volume 180, folio 66, subject to memoranda of mortgage 906006.7, 906006.8 and 186733.3.
  - (iv) The areas described in the Sixth Schedule shall be amalgamated with the land in certificate of title, Volume

- 462, folio 342, subject to memoranda of mortgage 906006.12, 186733.3 and 373862.1.
- (v) The areas described in the Seventh Schedule shall be amalgamated with the land in certificate of title, No. 33B/399.
- (vi) The areas described in the Eighth Schedule shall be amalgamated with the land in certificate of title, No. 33B/400.
- (vii) The areas described in the Ninth Schedule shall be amalgamated with the land in certificate of title, Volume 237, folio 33, subject to memorandum of mortgage 921789.1.
- (d) Declares the land described in the Tenth, Eleventh, Twelfth, Thirteenth and Fourteenth Schedules to be taken under section 119 (1) and declares that:
  - (i) The areas described in the Tenth Schedule shall be amalgamated with the land in certificate of title, No. 29D/892, subject to memorandum of mortgage 916768.13.
  - (ii) The area described in the Eleventh Schedule shall be amalgamated with the land in certificate of title, Volume 180, folio 66, subject to memoranda of mortgage 906006.7, 906006.8 and 186733.3.
  - (iii) The area described in the Twelfth Schedule shall be amalgamated with the land in certificate of title, Volume 462, folio 342, subject to memoranda of mortgage 906006.12, 186733.3 and 373862.1.
  - (iv) The areas described in the Thirteenth Schedule shall be amalgamated with the land in certificate of title, No. 33B/400.
  - (v) The area described in the Fourteenth Schedule shall be amalgamated with the land in certificate of title, No. 33B/399.

#### First Schedule

#### Wellington Land District

#### Land Acquired for Road

All those pieces of land situated in Block III, Hautapu Survey District, described as follows:

Area m²

<sup>2</sup> Being

- 91 Part Lot 3, D.P. 59354; coloured sepia on S.O. (3.6p) Plan 28285.
- 5 Part Lot 3, D.P. 59354; coloured sepia on S.O. (0.2p) Plan 28285.
- 476 Part Lot 3, D.P. 59354; coloured sepia on S.O. (18.8p) Plan 28285.
- 8 Part Lot 3, D.P. 59354; coloured sepia on S.O. (0.3p) Plan 28285.
- 33 Part Section 51; coloured orange on S.O. Plan (1.3p) 28285.
- 673 Part Section 9; coloured blue on S.O. Plan (26.6p) 28285.
- 473 Part Section 9; coloured blue on S.O. Plan (18.7p) 28285.
  - 81 Part Section 9; coloured blue on S.O. Plan
- (3.2p) 28285.
- 149 Part Section 35; coloured blue on S.O. Plan (5.9p) 28285.
- 680 Part Section 35; coloured blue on S.O. Plan (26.9p) 28286.

All those pieces of land situated in Block XV, Ohinewairua-Survey District described as follows:

Area m<sup>2</sup>

Being

1439 Part Lot 1, D.P. 65019; coloured orange on (0a1r16.9p) S.O. Plan 28285.

Area m<sup>2</sup> Being

225 Part Lot 1, D.P. 65019; coloured orange on (8.9p) S.O. Plan 28286.

5 Part Section 1; coloured orange on S.O. Plan (0.2p) 28286.

1692 Part Section 5; coloured sepia on S.O. Plan (0a1r26.9p) 28286.

As shown coloured as above mentioned on the plans, lodged in the office of the Chief Surveyor at Wellington.

#### Second Schedule

#### Wellington Land District

#### Land Declared to be Road

All those pieces of land situated in Block III, Hautapu Survey District, described as follows:

Area

 $m^2$ 

Being

2388 Part Lot 3, D.P. 59354; coloured sepia on S.O. (0a2r14.4p) Plan 28285.

1292 Part Section 9; coloured blue on S.O. Plan (0a1r11.1p) 28285.

2329 Part Section 10; coloured orange on S.O. Plan (0a2r12.1p) 28285.

8144 Part Section 10; coloured orange on S.O. Plan (2a0r02p) 28285.

3877 Part Section 35; coloured blue on S.O. Plan (0a3r33.3p) 28286.

All those pieces of land situated in Block XV, Ohinewairua Survey District, described as follows:

Area

m<sup>2</sup>

Being

1621 Part Lot 1, D.P. 65019; coloured orange on (0a1r24.1p) S.O. Plan 28286.

2157 Part Section 5; coloured sepia on S.O. Plan (0a2r5.3p) 28286.

1229 Part Section 5; coloured sepia on S.O. Plan (0a1r8.6p) 28286.

As shown coloured as above mentioned on the plans, lodged in the office of the Chief Surveyor at Wellington.

#### Third Schedule

#### Wellington Land District

Road Stopped and Amalgamated With Certificate of Title 29D/892

All those pieces of road situated in Block III, Hautapu Survey District, described as follows:

Area

m<sup>2</sup> Adjoining or passing through

261 Part Lot 3, D.P. 59354; coloured green on plan. (10.3p)

1614 Part Lot 3, D.P. 59354, and part Section 9; (0a1r23.8p) coloured green on plan.

440 Part Lot 3, D.P. 59354; coloured green on plan.

(17.4p) 1991 Part Lot 3, D.P. 59354, and part Section 10; (0a1r38.7p) coloured green on plan.

157 Part Lot 3, D.P. 59354; coloured green on plan.

(6.2p) 1482 Part Lot 3, D.P. 59354, and part Section 10; (0a1r18.6p) coloured green on plan.

As shown coloured as above mentioned on S.O. Plan 28285, lodged in the office of the Chief Surveyor at Wellington.

#### Fourth Schedule

#### Wellington Land District

Road Stopped and Amalgamated With Certificate of Title 170/243

All those pieces of road situated in Block III, Hautapu Survey District, described as follows:

Area m²

Adjoining or passing through

58 Part Section 9; coloured green on plan.

2.3p)

5 Part Section 9; coloured green on plan.

(0.2p)

81 Part Section 9; coloured green on plan.

(3.2p)

As shown coloured as above mentioned on S.O. Plan 28285, lodged in the office of the Chief Surveyor at Wellington.

#### Fifth Schedule

#### Wellington Land District

Road stopped and Amalgamated With Certificate of Title 180/66

All that piece of road containing 1199 square metres (0a1r07.4p), situated in Block III, Hautapu Survey District, adjoining or passing through part Section 10 and part Lot 3, D.P. 59354; coloured green on S.O. Plan 28285, lodged in the office of the Chief Surveyor at Wellington.

#### Sixth Schedule

#### Wellington Land District

Road Stopped and Amalgamated With Certificate of Title 462/342

All those pieces of road situated in Block III, Hautapu Survey District, described as follows:

Area

m<sup>2</sup> Adjoining or passing through

395 Part Section 35; coloured green on S.O. Plan (15.6p) 28285.

(15.6p) 28285.

1394 Part Section 35, Block III, Hautapu Survey (0a1r15.1p) District, and part Lot 1, D.P. 65019, Block XV, Ohinewairua Survey District; coloured green on S.O. Plan 28286.

As shown coloured as above mentioned on the plans, lodged in the office of the Chief Surveyor at Wellington.

#### Seventh Schedule

#### Wellington Land District

Road Stopped and Amalgamated With Certificate of Title 33B/399

All those pieces of road situated in Block XV, Ohinewairua Survey District; described as follows:

Area

m<sup>2</sup>

Adjoining or passing through

2772 Part Lot 1, D.P. 65019, Block XV, Ohinewairua (0a2r29.6p) Survey District, and part Section 10, Block III, Hautapu Survey District; coloured green, edged green on S.O. Plan 28285.

3 Part Lot 1, D.P. 65019; coloured green on S.O. (0.1p) Plan 28285.

3799 Part Lot 1, D.P. 65019; coloured green on S.O. (0a3r30.2p) Plan 28286.

As shown coloured as above mentioned on the plans, lodged in the office of the Chief Surveyor at Wellington.

ICL.

#### Eighth Schedule

#### Wellington Land District

Road Stopped and Amalgamated With Certificate of Title 33B/400

All those pieces of road situated in Block XV, Ohinewairua Survey District; described as follows:

m²

Adjoining or passing through

417 Part Section 1; coloured green on plan.

(16.5p)

4310 Part Sections 1, 2 and 5; coloured green on (1a0r10.4p)plan.

As shown coloured as above mentioned on S.O. Plan 28286, lodged in the office of the Chief Surveyor at Wellington.

#### Ninth Schedule

#### Wellington Land District

Road Stopped and Amalgamated With Certificate of Title

All those pieces of road situated in Block XV, Ohinewairua Survey District; described as follows:

Area

m² Adjoining or passing through

40 Part Section 5; coloured green on plan.

(1.6p)

1887 Part Sections 2 and 5; coloured green on plan.

(0a1r34.6p)

1576 Part Section 5; coloured green on plan.

(0a1r22.3p)

As shown coloured as above mentioned on S.O. Plan 28286, lodged in the office of the Chief Surveyor at Wellington.

#### Tenth Schedule

#### Wellington Land District

Land Taken and Amalgamated With Certificate of Title

All those pieces of land situated in Block III, Hautapu Survey District; described as follows:

Area

m<sup>2</sup>

Being

202 Part Section 9; coloured blue, edged blue on (8.0p)

501 Part Section 10; coloured orange, edged orange, (19.8p)on plan.

1055 Part Section 10; coloured orange, edged orange, (0a1r1.7p)on plan.

As shown coloured as above mentioned on S.O. Plan 28285, lodged in the office of the Chief Surveyor at Wellington.

#### Eleventh Schedule

#### Wellington Land District

Land Taken and Amalgamated With Certificate of Title 180/66

All that piece of land containing 18 square metres (0.7p), situated in Block III, Hautapu Survey District, being part Lot 3, D.P. 59354; coloured sepia, edged sepia, on S.O. Plan 28285, lodged in the office of the Chief Surveyor at Wellington.

#### Twelfth Schedule

#### Wellington Land District

Land Taken and Amalgamated With Certificate of Title 462/342

All that piece of land containing 10 square metres (0.4p), situated in Block XV, Ohinewairua Survey District, being part Lot 1, D.P. 65019; coloured orange, edged orange, on S.O. Plan 28286, lodged in the office of the Chief Surveyor at Wellington.

#### Thirteenth Schedule

#### Wellington Land District

Land Taken and Amalgamated With Certificate of Title 33B/400

All those pieces of land situated in Block XV, Ohinewairua Survey District, described as follows:

Area

Being

321 Part Section 5; coloured sepia, edged sepia, on (12.7p)plan.

476 Part Section 5; coloured sepia, edged sepia, on (18.8p)plan

As shown coloured as above mentioned on S.O. Plan 28286. lodged in the office of the Chief Surveyor at Wellington.

#### Fourteenth Schedule

#### Wellington Land District

Land Taken and Amalgamated With Certificate of Title 339/399

All those pieces of land situated in Block III, Hautapu Survey District; described as follows:

Area m<sup>2</sup>

Being

46 Part Section 35; coloured blue, edged blue, on (1.8p)S.O. Plan 28286.

1889 Part Section 10; coloured orange, edged orange, (0a1r34.7p) on S.O. Plan 28285.

As shown coloured as above mentioned on the plans, lodged in the office of the Chief Surveyor at Wellington.

Dated at Wellington this 31st day of August 1990.

E. C. MELDRUM, District Manager.

(DOSLI Wg. D.O. 44/19/0)

ln9953

#### Land Held for Public Works Set Apart for Buildings of the General Government

Pursuant to section 52 (1) of the Public Works Act 1981, and a delegation from the Minister of Lands, the District Manager, Department of Survey and Land Information, Napier, declares the land described in the Schedule to be set apart for buildings of the general government.

#### Schedule

#### Hawke's Bay Land District

4909 square metres, situated in the City of Napier, being Section 1, as shown on S.O. 10029, lodged in the office of the Chief Surveyor, Napier.

Dated at Napier this 7th day of September 1990.

J. A. TOBIN, District Manager.

(DOSLI Na. D.O. 5/170) In10075

#### Land Acquired for a Limited Access Road (State Highway No. 1) in the District of Rangitikei

Pursuant to section 20 of the Public Works Act 1981, section 88 (2) of the Transit New Zealand Act 1989 and to a delegation from the Minister of Lands, the District Solicitor, Department of Survey and Land Information, Wanganui, declares that, an agreement to that effect having been entered into, the land described in the Schedule hereto is hereby acquired for limited access road and has become road, limited access road and State highway and vested in the Crown on the date of publication hereof in the *Gazette*.

#### Schedule

#### Wellington Land District—Rangitikei District

5316 square metres, situated in Block XVI, Whangaehu Survey District, being part Lot 1, D.P. 7915; as shown marked "A" on S.O. Plan 36366, lodged in the office of the Chief Surveyor at Wellington.

Dated at Wanganui this 6th day of September 1990.

B. P. BONISCH, District Solicitor.

(DOSLI Wg. D.O. 8/1/0/63/1:695181) ln10076

1CL

#### Land in Tauranga District Acquired for Service Lane and Local Purpose Reserve (Service Lane)

Pursuant to section 20 of the Public Works Act 1981 and to a delegation from the Minister of Lands, the District Solicitor of the Department of Survey and Land Information, Hamilton, declares that, agreements to that effect having been entered into, the land described in the First Schedule hereto is hereby acquired for a service lane and the land described in the Second Schedule hereto is hereby acquired for local purpose reserve (service lane), and further declares that the land in both Schedules shall vest in The Tauranga District Council on the date of publication hereof in the *Gazette*.

#### First Schedule

#### South Auckland Land District

Area m²	Being
49	Part Lot 36, D.P. 18318; marked "A" on plan.
49	Part Lot 37, D.P. 18318; marked "B" on plan.
49	Part Lot 38, D.P. 18318; marked "C" on plan.
49	Part Lot 39, D.P. 18318; marked "D" on plan.
49	Part Lot 40, D.P. 18318; marked "E" on plan.
49	Part Lot 41, D.P. 18318; marked "F" on plan.
49	Part Lot 42, D.P. 18318; marked "G" on plan.
49	Part Lot 43, D.P. 18318; marked "H" on plan.
49	Part Lot 45, D.P. 18318; marked "K" on plan.
49	Part Lot 47, D.P. 18318; marked "M" on plan.
37	Part Lot 48, D.P. 18318; marked "N" on plan.
49	Part Lot 49, D.P. 18318; marked "O" on plan.
49	Part Lot 50, D.P. 18318; marked "P" on plan.
49	Part Lot 51, D.P. 18318; marked "Q" on plan.
49	Part Lot 52, D.P. 18318; marked "R" on plan.
49	Part Lot 53, D.P. 18318; marked "S" on plan.
49	Part Lot 54, D.P. 18318; marked "T" on plan.
49	Part Lot 55, D.P. 18318; marked "U" on plan.
48	Part Lot 56, D.P. 18318; marked "V" on plan.
47	Part Lot 58, D.P. 18318; marked "W" on plan.
83	Part Lot 60, D.P. 18318; marked "X" on plan.

Situated in Block VII, Tauranga Survey District.

As shown marked as above mentioned on S.O. Plan 58259, lodged in the office of the Chief Surveyor at Hamilton.

#### Second Schedule

#### South Auckland Land District

Area m²	Being
1	Part Lot 56, D.P. 18318; marked "AA" on plan.
49	Part Lot 44, D.P. 18318; marked "J" on plan.
1	Part Lot 58, D.P. 18318; marked "CC" on plan.
1	Part Lot 58, D.P. 18318; marked "DD" on plan.
49	Part Lot 46, D.P. 18318; marked "L" on plan.

Area m²

Being

1 Part Lot 60, D.P. 18318; marked "FF" on plan.

Situated in Block VII, Tauranga Survey District. As shown marked as above mentioned on S.O. Plan 58259, lodged in the office of the Chief Surveyor at Hamilton.

Dated at Hamilton this 7th day of September 1990.

W. G. KORVER, District Solictor.

(DOSLI Hn. 98/9/0/2)

ln10117

1CL

#### Land in Whakatane District Acquired for Soil Conservation and River Control Purposes

Pursuant to section 20 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor of the Department of Survey and Land Information, Hamilton, declares that, agreement to that effect having been entered into, the land described in the Schedule hereto is hereby acquired for soil conservation and river control purposes and shall vest in The Bay of Plenty Regional Council on the date of publication hereof in the *Gazette*.

#### Schedule

#### South Auckland Land District

2252 square metres, situated in Block III, Rangitaiki Upper Survey District, being Section 1, S.O. Plan 48620.

Dated at Hamilton this 7th day of September 1990.

W. G. KORVER, District Solictor.

(DOSLI Hn. 96/154000/1/0/13) ln10118

1CL

# Land Set Apart for Granting as Compensation in Whakatane District

Pursuant to section 52 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor, Department of Survey and Land Information, Hamilton, declares the land described in the Schedule hereto to be set apart for granting as compensation.

#### Schedule

Area

#### South Auckland Land District

71100				
m <sup>2</sup> Being				
2 Allotment 883, Matata Parish				
123 Allotment 885, Matata Parish				
271 Allotment 890, Matata Parish				
558 Allotment 891, Matata Parish				
Situated in Block V, Rangitaiki Upper Survey District.				
Dated at Hamilton this 6th day of Sep	tember 1990.			
W. G. KORVER, District Solicitor.				
(DOSLI Hn. D.O. 96/153000/2/0/2) 1CL				
ln10119				

#### Road Declared Stopped and Land Acquired for Soil Conservation and River Control Purposes in Hauraki District

Pursuant to Parts VIII and II of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor of the Department of Survey and Land Information, Hamilton:

- (a) Pursuant to section 116, declares the piece of road described in the First Schedule hereto to be stopped.
- (b) Pursuant to section 20, declares that, an agreement to that effect having been entered into, the said piece of stopped road described in the Second Schedule hereto is hereby acquired for soil conservation and river control purposes and

shall vest in The Waikato Regional Council on the date of publication hereof in the *Gazette*.

#### First Schedule

#### South Auckland Land District

All that piece of road containing 1020 square metres, adjoining or passing through Lots 2 and 3, D.P. S. 31391 and part Okere Block; as shown marked "I" on S.O. Plan 56212, lodged in the office of the Chief Surveyor at Hamilton.

#### Second Schedule

#### South Auckland Land District

All that piece of stopped road containing 1020 square metres, situated in Blocks XI and XII, Thames Survey District, now known as Section 1, S.O. Plan 56212.

Dated at Hamilton this 6th day of September 1990.

W. G. KORVER, District Solicitor.

(DOSLI Hn. D.O. 96/09200/4/0/289) in10120

1CL

#### Revoking a Declaration Acquiring Stopped Road for Soil Conservation and River Control Purposes in Hauraki District

Pursuant to section 55 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor of the Department of Survey and Land Information, Hamilton, revokes the declaration dated the 20th day of November 1987, published in the *New Zealand Gazette* of 3 December 1987, No. 214, page 5553, acquiring stopped road for soil conservation and river control purposes in Blocks XI and XII, Thames Survey District, pursuant to section 20 of the Public Works Act 1981 (registered at the Land Registry Office at Hamilton as H. 771283).

Dated at Hamilton this 6th day of September 1990.

W. G. KORVER, District Solicitor.

(DOSLI Hn. D.O. 96/092000/4/0/289)

1CL

# Land Set Apart for Pre-School Education Purposes in Hamilton City

Pursuant to section 52 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor, Department of Survey and Land Information, Hamilton, declares the land described in the Schedule hereto to be set apart for pre-school education purposes.

#### Schedule

#### South Auckland Land District

820 square metres, situated in Block IX, Komakorau Survey District, being Lot 152 on D.P. S. 56185. All certificate of title No. 46C/729.

Dated at Hamilton this 6th day of September 1990.

W. G. KORVER, District Solicitor.

(DOSLI Hn. 39/465/0) ln10122

1CL

# Land Set Apart for State Primary School in Hamilton City

Pursuant to section 52 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor, Department of Survey and Land Information, Hamilton, declares the land described in the Schedule hereto to be set apart for State primary school.

#### Schedule

#### South Auckland Land District

2.1200 hectares, situated in Block IX, Komakorau Survey District, being Lot 1, D.P. S. 56186. All certificate of title No. 46C/730.

Dated at Hamilton this 6th day of September 1990.

W. G. KORVER, District Solicitor.

(DOSLI Hn. 39/465/0) ln10123

1CL

#### Land Acquired for Road in Otorohanga District

Pursuant to section 20 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor of the Department of Survey and Land Information, Hamilton, declares that, agreement to that effect having been entered into, the land described in the Schedule hereto is hereby acquired for road and shall vest in The Otorohanga District Council on the date of publication hereof in the *Gazette*.

#### Schedule

#### South Auckland Land District

1615 square metres, situated in Block V, Wharepapa Survey District, being part Lot 3, D.P. S. 35952; as shown marked "A" on S.O. Plan 58488, lodged in the office of the Chief Surveyor at Hamilton.

Dated at Hamilton this 7th day of September 1990.

W. G. KORVER, District Solictor.

(DOSLI Hn. 98/6/0/60)

in10124

1Cl

# Land Acquired for Sewage Treatment Works in Chatton Survey District

Pursuant to section 20 (1) of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Manager, Department of Survey and Land Information, Invercargill, declares that, an agreement to that effect having been entered into, the land described in the Schedule hereto is hereby acquired for sewage treatment works and shall vest in The Gore District Council on the date of publication hereof in the Gazette.

#### Schedule

#### Southland Land District

All those pieces of land situated in Block XIV, Chatton Survey District, described as follows:

Area m²

Being

- 2344 Part Section 100 as shown marked "A" on plan. Part certificate of title B3/315.
- 1838 Part Section 101 as shown marked "B" on plan. Part certificate of title 151/82.
- 2622 Part Section 122 as shown marked "C" on plan. Part certificate of title 2A/1449.
- 659 Part Section 104 as shown marked "D" on plan. Part certificate of title 6C/702.

As shown as above mentioned on S.O. Plan 10924, lodged in the office of the Chief Surveyor at Invercargill.

Dated at Invercargill this 4th day of September 1990.

R. W. G. DALGLISH, District Manager.

(DOSLI In. 2100/P04)

ln9870

# Amending a Notice Setting Apart Land in Block VII, Tarawera Survey District

Pursuant to section 55 of the Public Works Act 1981, and to a delegation from the Minister of Lands; the District Manager.

Department of Survey and Land Information, Napier hereby amends the notice dated the 22nd day of May 1990, published in the New Zealand Gazette of the 31st day of May 1990, No. 87 at page 1955, setting apart land in Block VII, Tarawera Survey District by omitting the Schedule and substituting the following Schedule.

#### Schedule

#### Hawke's Bay Land District

All those pieces of land situated in Block VII, Tarawera Survey District, described as follows:

Area ha

Being

2.7222 Suburban Section 8, Tarawera.

 $m^2$ 

794 Part Town Section 32, Tarawera.

804 Part Town Section 33, Tarawera.

3246 Part Town Sections 21, 22, 23 and 24, Tarawera.

Dated at Napier this 3rd day of September 1990.

J. A. TOBIN, District Manager.

(DOSLI Na. D.O. 7/145) ln9871

#### Stopped Road in Kaipara District to be Vested

Pursuant to section 117 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Auckland, declares the part of stopped road described in the Schedule to be amalgamated with the land contained in certificate of title, Volume 915, folio 162, North Auckland Land Registry.

#### Schedule

#### North Auckland Land District

500 square metres being Section 1, S.O. Plan 60554, lodged in the office of the Chief Surveyor at Auckland. (Road stopped, Gazette notice C. 130474.1, North Auckland Land Registry).

Dated at Auckland this 3rd day of September 1990.

G. A. DAWSON, Manager (Lands and Property).

(DOSLI Ak. D.O. 50/15/12/0/60554) In9918

1CL

#### Declaring Land to be Road and Road Stopped in Far North District

Pursuant to Part VIII of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Auckland, declares:

- (a) Pursuant to section 114, the land described in the First Schedule to be road which shall vest in The Far North District
- (b) Pursuant to section 116, the portions of road described in the Second Schedule to be stopped.

#### First Schedule

#### North Auckland Land District

Area

 $m^2$ Being

453 Part Allotment 6, Mangonui East Parish; marked "A" on plan.

1380 Part Lot 2, D.P. 25929; marked "B" on plan.

925 Part Lot 2, D.P. 25929; marked "D" on plan.

Shown marked as above mentioned on S.O. Plan 64117, lodged in the office of the Chief Surveyor at Auckland.

#### Second Schedule

#### North Auckland Land District

Area  $m^2$ 

Adjoining or passing through

6000 Lots 2 and 3, D.P. 25929, marked "A" on plan. 9800 Lot 3, D.P. 25929 and part Allotment W17, Parish of Mahurangi East; marked "B" on plan.

Shown marked as above mentioned on S.O. Plan 63285, lodged in the office of the Chief Surveyor at Auckland.

Dated at Auckland this 3rd day of September 1990.

G. A. DAWSON, Manager (Lands and Property).

(DOSLI Ak. D.O. 50/15/16/0/63285 and 64117) ln9919

1CL

#### Land Acquired for Ngunguru Road in Whangarei District

Pursuant to section 20 (1) of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Auckland, declares that, agreements to that effect having been entered into, the land described in the Schedule is acquired for road which shall vest in The Whangarei District Council on the date of publication in the Gazette.

#### Schedule

#### North Auckland Land District

Area  $m^2$ Being

3735 Part Huanui Block; marked "A" on plan.

840 Part Huanui Block; marked "B" on plan.

2370 Part Huanui Block; marked "C" on plan. 60 Part Pukepoto A Block; marked "D" on plan.

385 Part Old Bed of Waitangi Stream; marked "J" on plan.

20 Part Old Bed of Waitangi Stream; marked "I" on

Shown marked as above mentioned on S.O. Plan 65323, lodged in the office of the Chief Surveyor at Auckland.

Dated at Auckland this 3rd day of September 1990.

G. A. DAWSON, Manager (Lands and Property).

(DOSLI Ak. D.O. 50/15/15/0/65323) ln9920

1CL

#### Declaring Road to be Stopped (Ngunguru Road), Whangarei District

Pursuant to section 116 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Auckland, declares the portions of road described in the Schedule to be stopped.

#### Schedule

#### North Auckland Land District

Area

 $m^2$ Adjoining or passing through

3310 Part Huanui Block; marked "F" on plan.

1135 Part Huanui Block; marked "G" on plan.

Shown marked as above mentioned on S.O. Plan 65323. lodged in the office of the Chief Surveyor at Auckland.

Dated at Auckland this 3rd day of September 1990.

G. A. DAWSON, Manager (Lands and Property).

(DOSLI Ak. D.O. 50/15/15/0/65323) ln9921

1CL

### Land Acquired for Taipa Heights Road, Far North District

Pursuant to section 20 (1) of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Auckland, declares that, an agreement to that effect having been entered into, the land described in the Schedule is acquired for road which shall vest in The Far North District Council on the date of publication in the Gazette.

#### Schedule

#### North Auckland Land District

843 square metres, being part Lot 2, D.P. 84608; shown marked "C" on S.O. Plan 62901, lodged in the office of the Chief Surveyor at Auckland.

Dated at Auckland this 3rd day of September 1990.

G. A. DAWSON, Manager (Lands and Property).

(DOSLI Ak. D.O. 50/15/16/0/62901)

1CL

### Land Acquired for Taipa Heights Road, Far North District

Pursuant to section 20 (1) of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Auckland, declares that, an agreement to that effect having been entered into, the land described in the Schedule is acquired for road which shall vest in The Far North District Council on the date of publication in the Gazette.

#### Schedule

#### North Auckland Land District

843 square metres, being part Lot 2, D.P. 84608; shown marked "C" on S.O. Plan 62901, lodged in the office of the Chief Surveyor at Auckland.

Dated at Auckland this 3rd day of September 1990.

G. A. DAWSON, Manager (Lands and Property).

(DOSLI Ak. D.O. 50/15/16/0/62901) ln9923

1CL

# Amending a Notice, Land Declared to be Road and Road Stopped in the Southland District

Pursuant to section 55 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Manager, Department of Survey and Land Information, Invercargill, hereby amends the notice dated 14 November 1986, published in the *New Zealand Gazette*, 20 November 1986, No. 185, at page 4942, by revoking it in its entirety and substituting it with the following:

Pursuant to Part VIII of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Manager, Department of Survey and Land Information, Invercargill:

- (a) Pursuant to section 114, declares the land described in the First Schedule hereto to be road and vested in The Southland District Council.
- (b) Pursuant to sections 116 and 117, declares the portions of road described in the Second Schedule hereto to be stopped and declares that:
  - (i) The areas marked 'C', 'G' and 'I' on S.O. Plan 9905, shall be amalgamated with the land in certificate of title No. 9C/729, subject to existing encumbrances.
  - (ii) The area marked 'D' on S.O. Plan 9905, shall be amalgamated with the land in certificate of title No. 5C/1060, subject to existing encumbrances.

#### First Schedule

#### Southland District

Land Declared to be Road

All those pieces of land situated in Takitimu Survey District, described as follows:

Area

ha Being

- 2.2461 Part Section 280, Block XIX; marked 'A' on S.O. Plan 9904.
- 0.1510 Part Run 611, Block XIX; marked 'D' on S.O. Plan 9904.
- 0.8182 Crown land, Block XIX; marked 'C' on S.O. Plan 9904.
- 0.0891 Part Lot 1, D.P. 9894, Block XIX; marked 'B' on S.O. Plan 9904.
- 0.3454 Part Section 280, Block XIX; marked 'A' on S.O. Plan 9905
- 1.5056 Part Section 289, Block XVIII; marked 'B' on S.O. Plan 9905.
- 0.5471 Part Section 280, Block XIX; marked 'E' on S.O. Plan 9905
- 0.9260 Part Section 286, Block XVIII; marked 'F' on S.O. Plan 9905.
- 0.4226 Part Section 287, Block XVIII; marked 'H' on S.O. Plan 9905.
- 0.0176 Part Section 286, Block XVIII; marked 'J' on S.O. Plan 9905.

As shown marked on the plans as above mentioned lodged in the office of the Chief Surveyor at Invercargill.

#### Second Schedule

#### Southland District

Road Stopped

All those pieces of road situate in Takitimu Survey District, described as follows:

Area ha

Adjoining or passing through

1CL

- 0.4468 Section 280, Block XIX; marked 'C' on plan.
- 1.5739 Section 280, Block XIX; marked 'G' on plan.
- 0.4886 Section 286, Block XVIII; marked 'I' on plan.
- 0.0655 Section 271, Block XVIII; marked 'D' on plan.

As shown marked as above mentioned on S.O. Plan 9905, lodged in the office of the Chief Surveyor at Invercargill.

Dated at Invercargill this 7th day of September 1990.

R. W. G. DALGLISH, District Manager.

(DOSLI In. 2100/P04)

ln10087

#### Amending a Notice Acquiring Land for Victoria University Purposes (Student Accommodation) in the City of Wellington

Pursuant to section 55 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Wellington, hereby amends the notice dated the 22nd day of December 1989 and published in the *New Zealand Gazette* of 18 January 1990, No. 5 at page 144, by omitting the following:

"and easement of light, air and overhanding eaves" and substituting the following:

"and appurtenant to easements of light, air and overhanging eaves"

Dated at Wellington this 4th day of September 1990.

D. I. GRAY, Manager, Lands and Property. (DOSLI Wn. D.O. PL 9/3/6)

13 SEPTEMBER

ln10092

1CL

Amending a Notice Declaring Road to be Stopped and Vested in the Natural Gas Corporation of New Zealand in Block XIV, Wairoa Survey District

Pursuant to section 55 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Wellington, hereby amends the notice dated the 8th day of August 1990 and published in the New Zealand Gazette of 1990, No. 141 at page 2914, by omitting the following:

"situated in Block XIV, Wairoa Survey District" and substituting the following:

"adjoining or passing through Section 143 Waitotara District, Lot 1 D.P. 59772 and Section 4 S.O. 34844"

Dated at Wellington this 3rd day of September 1990.

D. I. GRAY, Manager, Lands and Property.

(DOSLI Wn. D.O. PL 51/20/1)

1CL

#### Amending a Notice Declaring Road to be Stopped and Vested in the Crown for Railway Purposes in Block VIII, Paekakariki Survey District, Hutt County

Pursuant to section 55 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Wellington, hereby amends the notice dated the 7th day of August 1990 and published in the New Zealand Gazette of 1990, No. 141 at page 2916, by omitting the following:

"being all the land"

and substituting the following:

"adjoining or passing through Lots 1, 6 and 3, D.P. 31435, contained"  $\,$ 

Dated at Wellington this 3rd day of September 1990.

D. I. GRAY, Manager, Lands and Property.

(DOSLI Wn. D.O. PL 35/20/1)

1CL

#### Reserve for the Purposes of an Automatic-Telephone Exchange and Reserve for the Purposes of a Site for a Post Office. Set Apart for the Purposes of a Public Work in the City of Wellington

Pursuant to section 52 (1) of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Wellington, declares that the reserve for the purposes of an automatic telephone exchange described in the First Schedule hereto to be set apart for a public work and the reserve for the purposes of a site for a post office described in the Second Schedule hereto to be set apart for a public work.

#### First Schedule

#### Wellington Land District—City of Wellington

All that piece of land containing 869 square metres, more or less, being Part 2 of part Lot 1, Block VIII, D.P. 6174, contained in *New Zealand Gazette*, 1926, page 3304.

#### Second Schedule

#### Wellington Land District—City of Wellington

All that piece of land containing 535 square metres, more or less, being Part 1 of part Lot 1, Block VIII, D.P. 6174, contained in *New Zealand Gazette*, 1923, page 845.

Dated this 3rd day of September 1990.

D. I. GRAY, Manager, Lands and Property.

(DOSLI Wn. D.O. PL 19/6/2) ln10097

1CL

# Declaring Land to be Road in Block XVI, Belmont Survey District.

Pursuant to section 114 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Wellington, declares the land described in the Schedule hereto, to be road and vested in The Lower Hutt City Council.

#### Schedule

#### Wellington Land District—City of Lower Hutt

All that piece of land containing 67 square metres, more or less, being part certificate of title, Volume 20B, folio 291, shown marked 'A' on S.O. Plan 36281, lodged in the office of the Chief Surveyor at Wellington.

Dated this 31st day of August 1990.

D. I. GRAY, Manager, Lands and Property.

(DOSLI Wn. D.O. PL 42/19/3)

1CL

# Declaring Land to be Road in Block XI, Wairoa Survey District

Pursuant to section 114 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Wellington, declares the land described in the Schedule hereto, to be road and vested in The South Taranaki District Council.

#### Schedule

#### Wellington Land District

Area

m<sup>2</sup>

Being

7895 Section 1 on S.O. Plan 36152, contained in Proclamation No. B067086.4.

590 Section 28 on S.O. Plan 24723 contained in Proclamation No. B067086.3.

Dated at Wellington this 31st day of August 1990.

D. I. GRAY, Manager, Lands and Property.

(DOSLI Wn. D.O. PL 53/19/1) ln10102

1CL

# Road Stopped and Vested in the Adjoining Owner in Block XIV, Akatarawa Survey District, City of Upper Hutt

Pursuant to sections 116 (1), 117 (3) and 120 (3) of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Wellington, declares the road described in the Schedule hereto to be stopped and shall be vested in Bernard Manly Priston of Wellington, public servant, and amalgamated with certificate of title, Volume 5A, folio 826.

#### Schedule

#### Wellington Land District—City of Upper Hutt

All that piece of land containing 82 square metres, adjoining or passing through Lot 2, D.P. 28359, shown marked 'G' on S.O. Plan 36066, lodged in the Office of the Chief Surveyor at Wellington.

Dated this 31st day of August 1990.

D. I. GRAY, Manager, Lands and Property.

(DOSLI Wn. D.O. PL 1/8/13) ln10103

1CL

#### Land in Otorohanga District Acquired for Stream Diversion Purposes and in Connection with Stream **Diversion Purposes**

Pursuant to section 20 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor of the Department of Survey and Land Information, Hamilton, declares that, an agreement to that effect having been entered into, the land described in the First Schedule hereto is hereby acquired for stream diversion purposes and the land described in the Second Schedule hereto is hereby acquired in connection with stream diversion purposes, and further declares that the land in both Schedules shall vest in The Otorohanga District Council on the date of publication hereof in the Gazette

#### First Schedule

#### South Auckland Land District

Area

 $m^2$ Being

455 Part Otorohanga Q2D Block; coloured sepia on

506 Part Otorohanga Q2D Block; coloured sepia on plan.

Situated in Block IV, Orahiri Survey District.

As shown coloured as above mentioned on S.O. Plan 42341, lodged in the office of the Chief Surveyor at Hamilton.

#### Second Schedule

#### South Auckland Land District

Area  $m^2$ 

Being

607 Part Waitomo Stream Bed; coloured sepia on plan. 1037 Part Waitomo Stream Bed; coloured yellow on plan.

Situated in Block IV, Orahiri Survey District.

As shown coloured as above mentioned on S.O. Plan 42341, lodged in the office of the Chief Surveyor at Hamilton.

Dated at Hamilton this 6th day of September 1990.

W. G. KORVER, District Solicitor.

(DOSLI Hn. 98/6/059)

ln10125

1CL

#### Land Declared Road in Western Bay of Plenty **District**

Pursuant to section 114 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Assistant District Manager, Department of Survey and Land Information, Hamilton, declares the land described in the Schedule hereto to be road and vested in The Western Bay of Plenty District Council.

#### Schedule

#### South Auckland Land District

Area

m<sup>2</sup>

Being

983 Part Crown grant; marked "E" on plan.

8 Part Crown grant; marked "I" on plan.

3158 Part Crown grant; marked "N" on plan.

As shown marked as above mentioned on S.O. Plan 57110, lodged in the office of the Chief Surveyor at Hamilton.

Dated at Hamilton this 10th day of September 1990.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 98/10/0/4) In10126

1CL

#### Road Realignment in Waitomo District

Pursuant to Part VIII of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Assistant District Manager of the Department of Survey and Land Information,

- (a) Pursuant to section 114, declares the land described in the First Schedule hereto to be road and vested in The Waitomo District Council.
- (b) Pursuant to section 116, declares the pieces of road described in the Second Schedule hereto to be stopped.
  - (c) Pursuant to section 117, declares that:
  - (i) Areas "R" and "U", now known as Sections 1 and 2, S.O. 57152 respectively, shall be amalgamated with the land in certificate of title, Volume 293, folio 201, South Auckland Land Registry.
  - (ii) Area "C", now known as Section 2, S.O. 57146, shall be amalgamated with the land in certificate of title No. 45D/751, subject to statutory land charge H. 221467 and to memorandum of mortgage H. 709471, South Auckland Land Registry.
  - (iii) Areas "A" and "E", now known as Section 1, S.O. 57146 and Section 1, S.O. 57147 respectively, shall be amalgamated with the land in certificate of title No. 18A/1257, South Auckland Land Registry.

#### First Schedule

#### South Auckland Land District

Area m<sup>2</sup>

Being

4426 Part Section 32, Block VIII, Maungamangero Survey District; marked "B" on S.O. Plan 57146.

1228 Part Lot 1, D.P. 13254; marked "M" on S.O. Plan 57148.

6 Part Kinohaku East 3D3A Block; marked "I" on S.O. Plan 57153.

Situated in Block VIII, Maungamangero Survey District.

Area  $m^2$ 

Being 407 Part Kinohaku East 3D3A Block; marked "T" on S.O. Plan 57152.

450 Part Kinohaku East 3D3A Block; marked "Z" on S.O. Plan 57153.

1733 Part Kinohaku East 3D3A Block; marked "F" on S.O. Plan 57153.

Situated in Block V, Otanake Survey District.

As shown on the plans marked as above mentioned and lodged in the office of the Chief Surveyor at Hamilton.

#### Second Schedule

#### South Auckland Land District

Area m<sup>2</sup>

Adjoining

1479 Part Kinohaku East 3D3A Block; marked "R" on S.O. Plan 57152.

167 Part Kinohaku East 3D3A Block; marked "U" on S.O. Plan 57152.

Situated in Block V, Otanake Survey District.

Area m<sup>2</sup>

7810 Section 32, Block VIII, Maungamangero Survey District; marked "E" on S.O. Plan 57147.

1CL

1CL

Area m<sup>2</sup>

1927 Section 10, Block IV, Maungamangero Survey District; marked "C" on S.O. Plan 57146.

Area ha

1.1660 Sections 29 and 32, Block VIII, Maungamangero Survey District; marked "A" on S.O. Plan 57146.

As shown on the plans marked as above mentioned and lodged in the office of the Chief Surveyor at Hamilton.

Dated at Hamilton this 10th day of September 1990.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 98/3/0/43)

ln10127

#### Stopped Road in Otorohanga District Vested

Pursuant to section 117 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Assistant District Manager of the Department of Survey and Land Information, Hamilton, declares that the:

- (a) Piece of stopped road firstly described in the Schedule hereto shall be amalgamated with the land in certificate of title No. 8C/892, South Auckland Land Registry.
- (b) Piece of stopped road secondly described in the Schedule hereto shall be amalgamated with the land in certificate of title No. 10B/889, subject to memoranda of mortgage H. 966355.4 and H. 966355.5, South Auckland Land Registry.

#### Schedule

#### South Auckland Land District

 $m^2$ Being

333 Section 1, S.O. Plan 44073.

258 Section 2, S.O. Plan 44073. Dated at Hamilton this 10th day of September 1990.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 98/6/0/16) ln10128

#### Road Realignment in Waikato District

Pursuant to Part VIII of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Assistant District Manager of the Department of Lands, Hamilton:

- (a) Pursuant to section 114, declares the land described in the First Schedule hereto to be road and vested in The Waikato District Council.
- (b) Pursuant to sections 116 and 117, declares the pieces of road described in the Second Schedule hereto to be stopped, and that:
  - (i) Area "A", now known as Section 1, S.O. 58274, shall be amalgamated with the land in certificate of title, Volume 679, folio 84, South Auckland Land Registry.
  - (ii) Area "B", now known as Section 2, S.O. 58274, shall be amalgamated with the land in certificate of title, Volume 358, folio 258, South Auckland Registry.
  - (iii) Area "C", now known as Section 3, S.O. 58274, shall be amalgamated with the land in certificate of title, Volume 802, folio 116, South Auckland Land Registry.
  - (iv) Area "D", now known as Section 4, S.O. 58274, shall be amalgamated with the land in certificate of title, Volume 358, folio 259, South Auckland Land Registry.
  - (v) Area "E", now known as Section 5, S.O. 58274, shall be amalgamated with the land in certificate of title, Volume 358, folio 257, South Auckland Land Registry.

#### First Schedule

#### South Auckland Land District

Area m²	Being
54	Part Lot 61, D.P. 15817; marked "G" on plan.
37	Part Lot 62, D.P. 15817; marked "H" on plan.
33	Part Lot 63, D.P. 15817; marked "I" on plan.
22	Part Lot 64, D.P. 15817; marked "J" on plan.
8	Part Lot 65, D.P. 15817; marked "K" on plan.

Situated in Block XV, Rangiriri Survey District.

As shown marked as above mentioned on S.O. Plan 58274, lodged in the office of the Chief Surveyor at Hamilton.

#### Second Schedule

#### South Auckland Land District

Area		
$m^2$	Adjoining	
	Part Lot 61, D.P. 15817; marked "A" on	
43	Part Lot 62, D.P. 15817; marked "B" on	plan.
43	Part Lot 63, D.P. 15817; marked "C" on	plan.
39	Part Lot 64, D.P. 15817; marked "D" on	plan.
30	Part Lot 65, D.P. 15817; marked "E" on	plan.

Situated in Block XV, Rangiriri Survey District.

As shown marked as above mentioned on S.O. Plan 58274, lodged in the office of the Chief Surveyor at Hamilton.

Dated at Hamilton this 10th day of September 1990.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 14/1/0/5) 1CL ln10129

#### Land in Tauranga District Declared to be Road

Pursuant to section 114 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Assistant District Manager of the Department of Survey and Land Information, Hamilton, declares the land described in the Schedule hereto to be road and vested in The Tauranga District Council.

#### Schedule

#### South Auckland Land District

96 square metres, situated in Block IX, Tauranga Survey District, being part Lot 2, D.P. S. 100; as shown marked "A" on S.O. Plan 58487, lodged in the office of the Chief Surveyor at Hamilton.

Dated at Hamilton this 10th day of September 1990.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 98/9/0/6) ln10130

#### Stopped Road and Land Taken Vested in Otorohanga District

Pursuant to sections 117 and 119 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Assistant District Manager of the Department of Survey and Land Information, Hamilton, declares that the piece of stopped road and land taken described in the Schedule hereto shall be amalgamated with the land in certificate of title, Volume 1103, folio 217, subject to memorandum of mortgage H. 108010.2, South Auckland Land Registry.

#### Schedule

#### South Auckland Land District

1.9422 hectares being Section 39, Block XI, Kawhia North Survey District.

Dated at Hamilton this 10th day of September 1990.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 17/7/38) ln10131

1CL

# Land in Thames-Coromandel District Declared to be Road

Pursuant to section 114 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Assistant District Manager of the Department of Survey and Land Information, Hamilton, declares the land described in the Schedule hereto to be road and vested in The Thames-Coromandel District Council.

#### Schedule

#### South Auckland Land District

6277 square metres situated in Block IV, Whitianga Survey District being part Waiwawa Block; as shown marked "A" on S.O. Plan 57562, lodged in the office of the Chief Surveyor at Hamilton.

Dated at Hamilton this 10th day of September 1990.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 98/2/0/69)

1CL

# Declaring Land in Waitomo District to be Crown Land

Pursuant to section 42 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor of

the Department of Survey and Land Information, Hamilton, declares the land described in the Schedule hereto to be Crown land subject to the Land Act 1948.

#### Schedule

#### South Auckland Land District

Area m<sup>2</sup>

Being

8220 Part Uekaha A13C Block. All *Gazette* notice S. 486955.

726 Section 3, S.O. 45136, being all *Gazette* notice H. 836174, together with the appurtenant water easement acquired by *Gazette* notice H. 873504.

Situated in Block X, Orahiri Survey District.

Area ha

Being

5.0420 Section 1, S.O. 58571.

Situated in Blocks X and XI, Orahiri Survey District.

Dated at Hamilton this 6th day of September 1990.

W. G. KORVER, District Solicitor.

(DOSLI Hn. 6900-1543/1)

ln10133

ICL

# **Regulation Summary**

#### Notice Under the Acts and Regulations Publication Act 1989

Pursuant to the Acts and Regulations Publication Act 1989, notice is hereby given of the making of regulations as under:

Authority for Enactment	Title or Subject-matter	Serial Number	Date of Enactment	Price Code	Postage and Pack- aging
Accident Compensation Act 1982	Accident Compensation (Public Hospitals Costs) Regulations 1990	1990/231	10/9/90	6-BY	\$2.20
Accident Compensation Act 1982	Accident Compensation (Private Hospitals Costs) Regulations 1990	1990/232	10/9/90	5-BY	\$2.20
Accident Compensation Act 1982	Accident Compensation (Specialists Costs) Regulations (No. 2) 1990	1990/233	10/9/90	20-CY	\$3.70
Accident Compensation Act 1982	Accident Compensation (Anaesthetists Costs) Regulations 1990	1990/234	10/9/90	10-CX	\$3.40
Accident Compensation Act 1982	Accident Compensation (Psychiatrists, Psychologists, and Psychotherapists Costs) Regulations 1990	1990/235	10/9/90	5-BY	\$2.20
Accident Compensation Act 1982	Accident Compensation (Audiologists Costs) Regulations 1990	1990/236	10/9/90	5-BY	\$2.20
Accident Compensation Act 1982	Accident Compensation (Radiologists Costs) Regulations 1990	1990/237	10/9/90	10-CX	\$3.40
Accident Compensation Act 1982	Accident Compensation (Dental Specialists Costs) Regulations (No. 2) 1990	1990/238	10/9/90	10-CX	\$3.40
Accident Compensation Act 1982	Accident Compensation (Dentists Costs) Regulations (No. 2) 1990	1990/239	10/9/90	7-BY	\$2.20
Accident Compensation Act 1982	Accident Compensation (Laboratory Services Costs) Regulations 1990	1990/240	10/9/90	5-BY	\$2.20
Accident Compensation Act 1982	Accident Compensation (Hyperbaric Oxygen Treatment Costs) Regulations 1990	1990/241	10/9/90	4-BX	\$2.00
Accident Compensation Act 1982	Accident Compensation (Referred Treatments Costs) Regulations 1990	1990/242	10/9/90	6-BY	\$2.20

### **NEW ZEALAND GAZETTE**

Authority for Enactment	Title or Subject-matter	Serial Number	Date of Enactment	Price Code	Postage and Pack- aging
Accident Compensation Act 1982	Accident Compensation (Pharmaceutical Costs) Regulations 1990	1990/243	10/9/90	4-BX	\$2.00
Accident Compensation Act 1982	Accident Compensation (Prescribed Artificial Limbs, Aids, and Prosthetic Appliances Costs) Regulations 1990	1990/244	10/9/90	4-BX	\$2.00
Accident Compensation Act 1982	Accident Compensation (Damaged or Lost Artificial Limbs, Aids, Clothing, and Spectacles Costs) Regulations 1990	1990/245	10/9/90	4-BX	\$2.00
Transport Act 1962	Transport (Infringement Notices) Regulations 1990	1990/246	10/9/90	13-CX	\$3.40
Transport Act 1962	Transport (Vehicle Standards) Regulations 1990	1990/247	10/9/90	25-CY	\$3.70
Securities Act 1978	Securities Act (ANZ Banking Group (New Zealand) Limited) Exemption Notice (No. 5) 1990	1990/248	10/9/90	2-A	\$1.50
Securities Act 1978	Securities Act (The New Zealand Society for the Intellectually Handicapped (Incorporated))  Exemption Notice 1990	1990/249	11/9/90	2-A	\$1.50

#### Postage and Packaging Charge: Mail Orders

If two or more copies ordered, the remittance should cover the cash price and the maximum charge for the total value of purchases as follows:

Total Value of Purchases \$	Maximum Charge \$
\$12.00 and less	1.50
\$12.01 and greater	3.25

Copies can be bought or ordered by mail from Government Bookshops. Please quote title and serial numbers. Prices for quantities supplied on application.

Government Bookshops are located at Housing Corporation Building, 25 Rutland Street (Private Bag, C.P.O.), Auckland 1; 33 Kings Street, Frankton (P.O. Box 857), Hamilton; Head Office, Mulgrave Street (Private Bag), Wellington 1; 25–27 Mercer Street (Private Bag), Wellington 1; 159 Hereford Street (Private Bag), Christchurch 1; Cargill House, 123 Princes Street (P.O. Box 1104), Dunedin; Government Buildings, 1 George Street, Palmerston North; E.S.T.V. House, 4185 Queens Drive, Lower Hutt.

V. R. WARD, Government Printer.



